Author's response to reviews

Title: Understanding competing risks: a simulation point of view

Authors:

Arthur Allignol (arthur.allignol@fdm.uni-freiburg.de)
Martin Schumacher (ms@imbi.uni-freiburg.de)
Christoph Wanner (Wanner_C@medizin.uni-wuerzburg.de)
Christiane Drechsler (Drechsler_C@medizin.uni-wuerzburg.de)
Jan Beyersmann (jan.beyersmann@fdm.uni-freiburg.de)

Version: 5 Date: 27 May 2011

Author's response to reviews:

Dear editors,

In this last revision of the manuscript, an acknowledgement of our grant has been added. Furthermore, as the final manuscript does not contain section numbers, cross-referencing in now done using the section names instead of the section numbers.

Best regards,

Arthur Allignol